

Response to the reviewer's comments:

1. In the section of AI applications about stomach, authors can add more relevant research content and clinical experiments, such as gastric lesion recognition, etc.

Response-We add more relevant research content and clinical experiments in the section of AI applications about stomach.

2. It might be an overstatement that GI field with AI may reach no limit in the future. With the development of this field, the marginal effect is bound to become more and more obvious. The authors can expand more on how to better improve the effect of artificial intelligence applications.

Response-We emphasized how we can improve the applications of the effects of artificial intelligence